Author's response to reviews

Title: Children who were vaccinated, breast fed and from low parity mothers live longer: A community based case- control study in Jimma, Ethiopia

Authors:

Belaineh Girma (belaineh@gmail.com)
Yemane Berhane (yemaneberhane@ethionet.et)

Version: 6 Date: 1 March 2011

Author's response to reviews: see over
Date  March 1, 2011
Subject: submission of revised manuscript

Dear Editor of the Journal

Thank you very much for the constructive comments we received. We have tried to accommodate all the editorial comments and advice given by the team. All given comments and suggestions have been taken up and acted upon.

Best Regards,